**Author’s response to reviews**

**Title:** A phase I/II study of weekly nab-paclitaxel plus cisplatin in chemotherapy-naïve patients with advanced non-small-cell lung cancer

**Authors:**

Yoshihiro Hattori (hattori@hp.pref.hyogo.jp)

Yuko Kono (kono119@hp.pref.hyogo.jp)

Shoichi Itoh (sho1.itoh@hp.pref.hyogo.jp)

Takako Inoue (okuyama-ta@mc.pref.osaka.jp)

Yoshiko Urata (urata@hp.pref.hyogo.jp)

Yoshitaka Kawa (ohayoubird@gmail.com)

Rie Tohnai (tohnair@hyogo-cc.jp)

Toru Kumagai (kumagai-to@mc.pref.osaka.jp)

Kazumi Nishino (nisino-ka@mc.pref.osaka.jp)

Ryuji Uozumi (uozumi@kuhp.kyoto-u.ac.jp)

Satoshi Morita (smorita@kuhp.kyoto-u.ac.jp)

Shunichi Negoro (negoro@takarazuka-cst.jp)

Fumio Imamura (imamura-fu@mc.pref.osaka.jp)

Miyako Satouchi (satouchi@hp.pref.hyogo.jp)

**Version:** 3  **Date:** 27 Jan 2020

**Author’s response to reviews:**

Editor Comments:

1. We note that two authors have the same initials (YK). To distinguish between them in the Authors’ Contributions section please designate them as YK1 and YK2, with YK1 corresponding to Yuko Kono, chosen because this name is furthest up the author list.
We revised these points.

(Authors' contributions section, line 5, page 12)